Author’s response to reviews

Title: Trauma center accessibility for road traffic injuries in Hanoi, Vietnam

Authors:

Takashi Nagata (nagata.takashi@gmail.com)
Ayako Takamori (ayako529@gmail.com)
Yoshinari Kimura (yoshi@lit.osaka-cu.ac.jp)
Akio Kimura (akimura@imcj.hosp.go.jp)
Makoto Hashizume (mhashi@med.kyushu-u.ac.jp)
Shinji Nakahara (snakahara@marianna-u.ac.jp)

Version: 6 Date: 15 September 2011

Author’s response to reviews: see over
September 15, 2011

Journal of Trauma Management and Outcomes.

Dear Editors:

Enclosed please find the revision of the manuscript titled “Trauma center accessibility for road traffic injuries in Hanoi, Vietnam” in resubmission to the journal as Original Paper.

A. All authors have contributed to each of three manuscript preparation activities: conception/design and analysis/interpretation; writing; and approval of the final version and will take public responsibility for the content of the paper.

B. The content of this report has not been published, nor is it being considered elsewhere. No known conflicts of interest pertaining to authors are pertinent to the publication of this report.

C. I revised the manuscript according to the Independent Editorial Production Team’s comments.

D. This paper will be of interest to emergency medicine practitioner, injury epidemiologists, and emergency physicians concerned with trauma care in the middle income countries.

Please do not hesitate to contact me if you have any questions concerning the resubmission of this manuscript. I look forward to hearing the results of the review.

Sincerely,

Takashi Nagata, MD
Corresponding Author
(Present) Department of Emergency Medicine and Critical Care Center, Kyushu University Hospital, 3-1-1, Maidashi, Higashi-ku, Fukuoka-city, 812-8582, Japan
(Contact)
Home address: 729-7 Nishimachi, Kurume, 830-0038, Japan
Phone & Fax (81)-942-325100
Email: nagata.takashi@gmail.com